Author's response to reviews

Title: Group versus individual sessions delivered by a physiotherapist for female urinary incontinence: an interview study with women attending group sessions nested within a randomised controlled trial

Authors:

Frances Griffiths (F.E.Griffiths@warwick.ac.uk)
Jo Pepper (jopepper@supanet.com)
Ellen C Jorstad-Stein (ellen.jorstad-stein@temple.edu)
Jan Fereday-Smith (Jan.Fereday-Smith@swh.nhs.uk)
Lesley Hill (Lesley.hill@geh.nhs.uk)
Sarah (Sallie) E Lamb (s.lamb@warwick.ac.uk)

Version: 5 Date: 4 September 2009

Author's response to reviews: see over
6th August 2009

The Editorial Office
BMC Women’s Health

Dear Sir, Madam,

Please find attached the revised manuscript for the incontinence study.

This paper relates to the Randomised Controlled Trial and is one of a pair. The other paper ("Group versus individual sessions delivered by a physiotherapist for female urinary incontinence: an interview study with women attending group sessions nested within a randomised controlled trial", reference MS2145945658206519) has been returned and is in the process of being published. We would like both papers to appear as a pair.

Please accept our sincere apologies for the delay in returning this paper. Unfortunately, we had to wait for the economist who was on long-term sick leave.

We look forward to hearing from BMC Women's Health.

Yours faithfully,

Professor Sallie Lamb
DPhil, MSc, MCSP, SRP
Professor of Rehabilitation
Director Clinical Trials Unit
Warwick Medical School Clinical Trials Unit
The University of Warwick
Gibbet Hill Road
Coventry
CV4 7AL